**f** 



# **Geophysical Research Letters**<sup>\*</sup>



10.1029/2021GL097627

#### **Key Points:**

- Unlike the sea-ice melting system, enhanced bacterial production (BP) in the eastern Getz Ice Shelf (GtzIS) was uncoupled with phytoplankton biomass
- Dissolved organic carbon in glacier meltwater likely stimulates BP on the eastern GtzIS
- Warming-induced glacial melting may weaken carbon sequestration in the Antarctic Ocean by stimulating heterotrophic bacterial metabolism

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J.-H. Hyun, hyunjh@hanyang.ac.kr

#### Citation:

Min, J.-O., Kim, S.-H., Jung, J., Jung, U.-J., Yang, E. J., Lee, S.H., & Hyun, J.-H. (2022). Glacial ice melting stimulates heterotrophic prokaryotes production on the Getz Ice Shelf in the Amundsen Sea, Antarctica. *Geophysical Research Letters*, 49, e2021GL097627. https://doi.org/10.1029/2021GL097627

Received 4 JAN 2022 Accepted 19 SEP 2022

#### **Author Contributions:**

Conceptualization: Jun-Oh Min, Sung-Han Kim, Eun Jin Yang, Jung-Ho Hyun Formal analysis: Jun-Oh Min, Sung-Han Kim, Jinyoung Jung, Ui-Jung Jung Funding acquisition: SangHoon Lee Investigation: Ui-Jung Jung Project Administration: SangHoon Lee Resources: Jinyoung Jung Supervision: Jung-Ho Hyun Visualization: Jun-Oh Min Writing – original draft: Jun-Oh Min, Jung-Ho Hyun

© 2022. The Authors. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

#### MIN ET AL.

# **Glacial Ice Melting Stimulates Heterotrophic Prokaryotes Production on the Getz Ice Shelf in the Amundsen Sea, Antarctica**

Jun-Oh Min<sup>1,2</sup>, Sung-Han Kim<sup>3</sup>, Jinyoung Jung<sup>2</sup>, Ui-Jung Jung<sup>3</sup>, Eun Jin Yang<sup>2</sup>, SangHoon Lee<sup>2</sup>, and Jung-Ho Hyun<sup>1</sup>

<sup>1</sup>Department of Marine Sciences and Convergent Technology, Hanyang University, Ansan, Republic of Korea, <sup>2</sup>Division of Ocean Sciences, Korea Polar Research Institute, Incheon, Republic of Korea, <sup>3</sup>Marine Environmental Research Center, Korea Institute of Ocean Science & Technology, Busan, Republic of Korea

**Abstract** Extensive oceanographic data sets were combined with microbiological parameters to elucidate the tight coupling between glacial meltwater and heterotrophic bacterial production (BP) on the Getz Ice Shelf (GtzIS) in the Amundsen Sea. BP in the eastern GtzIS (EG; 85.8 pM Leu.  $h^{-1}$ ), where basal glacier meltwater upwells, was significantly higher than BP measured in the western GtzIS (WG; 50.6 pM Leu.  $h^{-1}$ ) and the Amundsen Sea Polynya (ASP; 27.8 pM Leu.  $h^{-1}$ ). BP in the EG accounted for 49% of primary production, which was greater than that of the WG (10%) and ASP (9.2%). Enhanced BP in the eastern GtzIS was not coupled with phytoplankton biomass, but correlated significantly with the freshwater fraction containing meltwater-derived dissolved organic carbon (MW-DOC). These results suggest that warming-induced glacier melting weakens carbon sequestration efficiency in Antarctic coastal waters by stimulating heterotrophic metabolism that converts MW-DOC to CO<sub>2</sub>.

**Plain Language Summary** The Getz Ice Shelf (GtzIS) in the Amundsen Sea, the third-largest ice shelf (34,018 km<sup>2</sup>) on the West Antarctic Ice Sheet, is regarded as a hot spot for producing large amount of glacier meltwater ( $144.9 \pm 14$  Gt yr<sup>-1</sup>). In general, the amount of dissolved organic carbon (DOC) from terrestrial sources is negligible in Antarctic waters, making DOC supplied by phytoplankton a major carbon source for heterotrophic bacteria growth. Although Antarctica's glacial ice shelves are presumably a large reservoir of organic carbon (ca. 5.5 Pg C), little is known about the effect of glacial melting on the supply of DOC available to heterotrophic bacteria. On the eastern GtzIS, where glacial meltwater upwells to the surface water column, heterotrophic bacteria production has no significant positive relationship with phytoplankton biomass, but is positively correlated with freshwater fraction. This suggests that DOC in glacial meltwater, rather than phytoplankton, provides DOC to support enhanced bacteria production. Our results imply that global warming–related increases in glacial meltwater may stimulate heterotrophic bacterial metabolism that respires DOC to CO<sub>2</sub>, thereby reducing carbon sequestration efficiency in Antarctic coastal waters.

## 1. Introduction

Heterotrophic prokaryotes (hereafter bacteria) are a significant biological component that are almost exclusively responsible for the mineralization of dissolved organic carbon (DOC), the largest organic carbon pool in the ocean (Azam & Malfatti, 2007; Hansell & Carlson, 2015; Karl, 2007; D. L. Kirchman, 2008). DOC derived from various sources (Carlson & Hansell, 2015; Nagata, 2008) is either transferred to higher trophic levels or mineralized to  $CO_2$  via bacterial production (BP) and respiration, respectively, within the microbial loop (Azam et al., 1983; del Giorgio et al., 1997; Duarte & Prairie, 2005; Ducklow et al., 1986. Consequently, the relative importance of heterotrophic bacteria in the microbial loop ultimately regulates the efficiency of the biologically mediated carbon sequestration process in the ocean (Chisholm et al., 2001; Karl, 2007; Legendre & Le Fèvre, 1995).

Antarctic glaciers store huge amounts of organic carbon (ca. 5.5 Pg C), and the amount of organic carbon discharged through subglacial Antarctic meltwater is estimated to be 0.33 Tg C year<sup>-1</sup> as particle organic carbon (POC) and 0.17 Tg C year<sup>-1</sup> as DOC (Wadham et al., 2019). Organic carbon in glacier meltwater consists of substantial amounts of bioavailable components (Hood et al., 2015; Musilova et al., 2017; Smith et al., 2017). Recent climate change and the consequent rise in water temperatures have stimulated the melting of glaciers, resulting in a total glacial mass loss of 252 Gt per year between 2009 and 2017 (Rignot et al., 2019). As the



Writing – review & editing: Eun Jin Yang, Jung-Ho Hyun Southern Ocean accounts for approximately 20–50% of atmospheric  $CO_2$  uptake in the global ocean (Arrigo et al., 2008; Takahashi et al., 2009), it is important to elucidate the metabolic response of heterotrophic bacteria to increased supplies of DOC derived from glacier meltwater to construct models of biogeochemical carbon cycles in the Southern Ocean (Ducklow et al., 2012; Lønborg et al., 2018, 2019; Paulsen et al., 2017).

The Getz Ice Shelf (GtzIS) in the Amundsen Sea is the third-largest ice shelf (34,018 km<sup>2</sup>) on the West Antarctic Ice Sheet after the Ronne Ice Shelf (338,887 km<sup>2</sup>) and the Ross West Ice Shelf (306,105 km<sup>2</sup>) (Rignot et al., 2013). The GtzIS alone supports glaciers that hold enough ice to increase sea levels by 22 cm (Rignot et al., 2019). As a result of the inflow of warm Circumpolar Deep Water (CDW) into the west Antarctic shelf, including the Amundsen Sea (S. S. Jacobs et al., 2011; Jenkins et al., 2010), the GtzIS is now regarded as a hot spot that produces large amounts of glacier meltwater (144.9  $\pm$  14 Gt yr<sup>-1</sup>) (Morlighem et al., 2020; Rignot et al., 2013, 2019). Fresh and buoyant water released from a glacier cavity can drive regional and global ocean circulation (Jourdain et al., 2017; Nakayama et al., 2014; Raiswell et al., 2006; Silvano et al., 2018), while providing organic and inorganic substrates that stimulate biological production (Arrigo et al., 2017; Hood et al., 2015; Hopwood et al., 2018, 2020). Thus, increases in glacial meltwater can lead to changes in production and the community structure of phytoplankton (Arrigo et al., 2017; Alderkamp et al., 2012), as well as bacterial dynamics (Nicholes et al., 2019; Smith et al., 2017). Given that heterotrophic bacteria are a major consumers of DOC, enhanced bacterial metabolic activity associated with meltwaters may weaken carbon sequestration in Antarctic coastal waters (Ducklow et al., 2015; S. Lee et al., 2017). Despite the potential significance of meltwater-induced DOC release on bacterial dynamics, most microbiological oceanographic studies conducted in the water column of the Amundsen Sea Polynya (ASP) have been limited to bacterial dynamics coupled with various stages of phytoplankton blooms (Delmont et al., 2014; Hyun et al., 2016; J. G. Kim et al., 2014; S. J. Kim et al., 2019; B. Kim et al., 2022; Williams et al., 2016; Yager et al., 2012). Little information is available on the impact of glacier meltwater on the microbial metabolic activity in the Amundsen Sea, especially along the glacier ice shelves, largely due to an extensive sea-ice coverage making access difficult.

During an oceanographic research expedition to the Amundsen Sea in January and February 2016, we encountered a unique environmental condition in which most of the GtzIS was exposed to seawater (Figure 1). This provided us with direct opportunities to explore the bacterial dynamics associated with glacier meltwater input on the GtzIS. Here, we report that enhanced heterotrophic BP in front of the GtzIS was not coupled with phytoplankton that is supposedly the direct carbon source for heterotrophic bacteria in the Southern Ocean, but was directly related to glacier meltwater released from a basal part of the GtzIS affected by the intrusion of warm CDW. Our results are, to our knowledge, the first to demonstrate the importance of heterotrophic microbial processes in biogeochemical carbon cycles associated with glacier meltwater–induced DOC release along the west coast of Antarctica, where rapid glacial melting induced by climate change is occurring.

## 2. Materials and Methods

#### 2.1. Study Area

The Amundsen Sea is located along Marie Byrd Land between the Bellingshausen Sea and the Ross Sea. The GtzIS extends 650 km along the Pacific sector of the Antarctic coastline between 115° W and 135° W, in the central and western Amundsen Sea (Figure 1). West Antarctica, including the Amundsen Sea and Bellingshausen Sea, has undergone rapid warming (Brown et al., 2019; Montes-Hugo et al., 2009) in recent decades, with the surface water temperature increasing by approximately 1.5°C since the 1950s (Turner et al., 2014). The estimated basal melt rate on the GtzIS was an area-averaged 4.15 m year<sup>-1</sup>, resulting in a freshwater flux of 141.17 Gt year<sup>-1</sup> into the Southern Ocean (Wei et al., 2020). The glacier meltwaters discharged from the ice shelves in the Amundsen Sea during the summer of January 2011 and February 2012 reportedly extends more than 300 km offshore (I. Kim et al., 2016).

As a part of physical and biogeochemical processes studies in the Amundsen Sea, a multidisciplinary oceanographic survey was conducted onboard the RV Araon, a Korean research icebreaker, during the austral summer from January 14 to 16 February 2016. A temperature-salinity (T-S) diagram shows that the coastal waters of the Amundsen Sea can be divided into three main water masses: Antarctic Surface Water (AASW), Winter Water (WW), and modified Circumpolar Deep Water (mCDW) (S. S. Jacobs et al., 2011, Jenkins et al., 2010; Randall-Goodwin et al., 2015) (Figure S1 in Supporting Information S1). Water samples for microbiological

# **Geophysical Research Letters**

#### 10.1029/2021GL097627





Figure 1. A map of the study area with the sampling locations in the Amundsen Sea during January 14 to 16 February 2016. The contours indicate bathymetric lines in meters.

analyses were collected at nine stations along the GtzIS and in the center of the ASP (Figure 1 and Table S1 in Supporting Information S1). A detailed description of the sample collection and handling procedure is presented in the Supporting Information (Text S1 in Supporting Information S1).

Sampling sites were grouped into three regions: eastern GtzIS (EG), western GtzIS (WG) and ASP, according to the distribution of potential temperature, salinity, chlorophyll-a (Chl-a) concentration, and freshwater fraction (FWF) (Table 1 and Figure S1 in Supporting Information S1). The EG site was characterized by a higher basal glacier melting rate because the bottom ice depth was deeper than the thermocline depth and an ice cavity was in direct contact with warmer mCDW, whereas a low basal melting rate was observed in the WG because the relatively cold water was in contact with the ice cavity, and the shallow ice-bottom depth prevents warm mCDW from entering the WG cavity (e.g., Wei et al., 2020).

The surface waters of the WG site were characterized by the lowest temperature and salinity and highest FWF (Figures 2h and 2i), whereas the EG site exhibited the highest temperature and an intermediate salinity ranging between the salinity values observed in the ASP and the WG (Figures 2a and 2b). The ASP sites were characterized by high Chl-a concentrations with relatively greater thermal stratification compared with the GtzIS sites (Figures 2o and 2r).

Based on a satellite image of the ice distribution and the sharp density gradient, the surface waters of the WG were largely affected by sea-ice meltwater (Figures 1 and 2j). In contrast, the vertically well-mixed water column with high temperature and low salinity at the EG site indicates that warm mCDW containing glacier meltwater upwells along the ice shelf, resulting in a moderately low salinity range (Figures 2a–2c).

#### Table 1

Averaged (±1 SD) Physico-Chemical and Microbiological Parameters Within the Mixed Layer Depth in the Amundsen Sea

Oceanographic setting	Station	Depth (m)	Temp. (°C)	Sal. (psu)	Density (kg m <sup>-3</sup> )	Z <sub>eu</sub> (m)	MLD (m)	FWF (%)	DOC (µM)	Chl-a (µg L <sup>-1</sup> )	$BA \\ (\times 10^7 \text{ cells } L^{-1})$	BP (pM Leu. h <sup>-1</sup> )	BP:PP* (%)
Eastern Getz Ice Shelf	EG-1	674	0.28 (±0.02)	33.9 (±0.03)	27.2 (±0.03)	28.7	95.0	2.25 (±0.11)	49 (±2)	0.67 (±0.14)	43.2 (±3.32)	73.5 (±14.1)	49.8
	EG-2	1119	0.30 (±0.10)	33.9 (±0.03)	27.2 (±0.03)	28.0	53.4	2.50 (±0.10)	52 (±1)	0.63 (±0.18)	41.4 (±10.5)	79.9 (±17.6)	39.5
	EG-3	658	0.39 (±0.01)	33.9 (±0.04)	27.2 (±0.03)	26.0	25.7	2.44 (±0.13)	51 (±2)	0.77 (±0.03)	58.8 (±2.94)	104 (±4.14)	57.3
	Average (±1 SD)	817 (±262)	0.32 (±0.07)	33.9 (±0.01)	27.2 (±0.01)	27.6 (±1.40)	58.0 (±34.9)	2.40 (±0.13)	51 (±2)	0.69 (±0.07)	47.8 (±9.56)	85.8 (±16.1)	48.9 (±8.94)
Western Getz Ice Shelf	WG-1	596	-1.17 (±0.03)	33.4 (±0.03)	26.8 (±0.03)	24.0	32.6	4.20 (±0.12)	47 (±1)	1.79 (±0.54)	50.7 (±7.52)	50.4 (±3.15)	13.9
	WG-2	780	-1.11 (±0.05)	33.5 (±0.02)	27.0 (±0.02)	19.0	24.7	4.11 (±0.09)	46 (±2)	3.86 (±0.02)	49.3 (±1.89)	46.4 (±2.67)	7.11
	WG-3	323	-1.21	33.4 (±0.02)	27.0 (±0.02)	16.0	20.7	2.71 (±0.08)	51 (±1)	3.69 (±0.23)	53.9 (±9.86)	54.9 (±4.55)	9.48
	Average (±1SD)	566 (±230)	-1.16 (±0.05)	33.4 (±0.09)	26.9 (±0.07)	19.7 (±4.04)	26.0 (±6.06)	3.67 (±0.83)	48 (±3)	3.11 (±1.15)	51.3 (±2.36)	50.6 (±4.27)	10.2 (±4.98)
Amundsen Sea Polynya	ASP-1	697	0.26 (±0.02)	34.0	27.3	16.0	27.0	1.81	44 (±1)	3.00 (±0.81)	8.58 (±4.63)	26.8 (±3.56)	4.09
	ASP-2	820	-0.14 (±0.13)	34.0 (±0.02)	27.3 (±0.02)	16.0	64.0	1.96 (±0.07)	46 (±1)	2.17 (±0.94)	11.1 (±5.24)	28.7 (±7.90)	19.1
	ASP-3	700	-0.07 (±0.17)	34.0 (±0.02)	27.3 (±0.02)	22.8	44.0	2.07 (±0.06)	48 (±2)	1.62 (±0.47)	9.08 (±1.51)	29.9 (±7.54)	4.29
	Average (±1SD)	739 (±70)	-0.02 (±0.11)	34.0 (±0.10)	27.3 (±0.01)	18.1 (±4.00)	45.0 (±18.5)	1.95 (±0.13)	46 (±2)	2.26 (±0.70)	9.59 (±1.34)	27.8 (±0.92)	9.15 (±8.59)

Note. Z<sub>eu</sub>, euphotic depth; MLD, mixed layer depth; FWF, freshwater fraction; DOC, dissolved organic carbon; Chl-a, chlorophyll-a; BA, bacterial abundance; BP, bacterial production.

<sup>a</sup>BP:PP data were depth-integrated down to the MLD. Primary Production (PP) data from Lim et al. (2019).

#### 3. Results and Discussion

#### 3.1. Physico-Chemical Parameters

The FWF of the EG (range, 2.25%-2.44%; average,  $2.40 \pm 0.13\%$ ) displayed an intermediate range between those in the ASP (range, 1.81%-2.07%; average,  $1.95 \pm 0.13\%$ ) and WG (range, 2.71%-4.20%; average,  $3.67 \pm 0.83\%$ ) (Table 1). The warm mCDW intruding along the deep Dotson-Getz trough contacts the bottom of the ice shelf, thereby stimulating basal glacier melting in the EG (e.g., Wei et al., 2020). The vertically well-mixed water column with a deeper mixed-layer depth (MLD) (58  $\pm$  34.9 m) at the EG (Figures 2c and Table 1) suggests that upwelling of mCDW containing buoyant basal glacier meltwater destabilizes the water column (S. Jacobs et al., 2012; Mankoff et al., 2012). In the WG, where the FWF was highest (Table 1), low potential temperatures ranging from -1.2 to  $-1.11^{\circ}$ C (average,  $-1.16 \pm 0.05^{\circ}$ C) and salinity ranging between 33.4 and 33.5 psu (average,  $33.4 \pm 0.09$  psu) were observed within MLD (Table 1). Satellite images showed that sea-ice covering >50% of the sea surface was distributed widely in the WG (Figure 1). These results indicate that sea ice-derived meltwater in the WG contributed to the high FWF. The salinity gradient caused by the addition of low-salinity meltwater discharged from the sea ice was responsible for the formation of a strong pycnocline at relatively depths within 30–50 m of the WG (Figure 2j). At the ASP site, the salinity ranged from 34.0 to 34.2 psu, and the temperature ranged from -1.64 to 0.28°C (Figures 20 and 2p). The water in the upper 100 m of the ASP exhibited the properties of AASW (S = <34.1 psu, T = -1.80 to  $>0^{\circ}$ C) (Randall-Goodwin et al., 2015) that is fresher and warmer than the WW formed at 100–200 m due to sea-ice melting and surface heating by solar radiation.



# **Geophysical Research Letters**



Figure 2. Vertical profiles of physico-chemical and microbiological parameters in the Amundsen Sea.

In the present study, the NO<sub>2</sub><sup>-</sup> + NO<sub>3</sub><sup>-</sup> (11.75–31.98  $\mu$ M; average, 22.74 ± 6.13  $\mu$ M) and PO<sub>4</sub><sup>3-</sup> concentrations (0.94–2.20  $\mu$ M; average, 1.67 ± 0.40  $\mu$ M) (Table S1 in Supporting Information S1) appeared to be greater than the values reported in early summer from December 2013 to January 2014 (NO<sub>2</sub><sup>-</sup> + NO<sub>3</sub><sup>-</sup>, 3.4–34.6  $\mu$ M; PO<sub>4</sub><sup>3-</sup>, 0.8–2.3  $\mu$ M) (B. K. Kim et al., 2018). The concentrations of DOC in the study area ranged from 41 to 52  $\mu$ M (average, 47 ± 3  $\mu$ M; Figures 2g, 2n and 2u), and were similar to those reported at MLD during the austral summer in the Ross Sea (average, 46 ± 2 to 56 ± 5  $\mu$ M) (Carlson et al., 2000). The average DOC concentration at MLD was significantly higher in the EG (range, 49–52  $\mu$ M; average, 51 ± 2  $\mu$ M) than in the ASP (range, 44–48  $\mu$ M; average, 46 ± 2  $\mu$ M) and WG (range, 46–51; average, 48 ± 3  $\mu$ M) (Table 1). Relatively higher DOC concentrations at the EG sites were likely associated with upwelling of buoyant basal glacier meltwater containing abundant DOC (average, 51 ± 2  $\mu$ M) (Figure 2g and Figure S2b in Supporting Information S1).

#### 3.2. Phytoplankton Biomass Associated With the Water Column Structure

Concentrations of Chl-a were higher in the WG (average,  $3.11 \pm 1.15 \ \mu g \ L^{-1}$ ) than in the ASP (average,  $2.26 \pm 0.70 \ \mu g \ L^{-1}$ ) and EG (average,  $0.69 \pm 0.07 \ \mu g \ L^{-1}$ ) (Table 1). The significantly low Chl-a concentration in the EG (Kruskal–Wallis, p < 0.01) (Figure S2a in Supporting Information S1) was associated with the deep MLD, which resulted in a decrease in the residence time of phytoplankton in the euphotic layer. On the other hand, relatively high Chl-a concentrations in the WG were associated with the pycnocline formed by the salinity gradient in the shallow water column (30–50 m depth) due to the inflow of sea-ice meltwater (Figures 2i–2k), which increased the residence time of phytoplankton in the surface water column with sufficient light and nutrients to support high phytoplankton biomass. The Chl-a concentration in the ASP (average,  $2.26 \pm 0.70 \ \mu g \ L^{-1}$ ) (Table 1 and Figure 2r) was lower than the values reported in December 2013 to January 2014 (average,  $8.89 \pm 1.11 \ \mu g \ L^{-1}$ ) (Y. Lee et al., 2016) because phytoplankton blooms started further northeast of the ASP and the sampling sites at ASP along our sampling transect were not characterized by the blooms at the time of sampling (Fang et al., 2020).

#### 3.3. Enhanced Bacterial Production Uncoupled With Phytoplankton in the Glacier Melting System

Unlike Chl-a, BP was significantly greater in the EG (range, 73.5–104 pM Leu. h<sup>-1</sup>; average, 85.8 ± 16.1 pM Leu. h<sup>-1</sup>) with deep mixed-layer depth (MLD) than in the WG (range, 46.4–54.9 pM Leu. h<sup>-1</sup>; average, 50.6 ± 4.27 pM Leu. h<sup>-1</sup>) and ASP (range, 26.8–29.9 pM Leu. h<sup>-1</sup>; average, 27.8 ± 0.92 pM Leu. h<sup>-1</sup>) (Kruskal–Wallis, p < 0.01; Figure S2d in Supporting Information S1). In Antarctic waters where terrestrial dissolved organic matter (DOM) input is negligible (Ducklow et al., 2007), DOM supplied by phytoplankton is a major source of carbon supporting heterotrophic bacterial growth (Ducklow et al., 2012; H. Kim and Ducklow, 2016; Morán et al., 2001; Ortega-Retuerta et al., 2008). Previous studies in the ASP revealed that BP is tightly coupled with phytoplankton biomass (Hyun et al., 2016; Williams et al., 2016). In the present study, the degree of coupling between BP and Chl-a varied according to spatial variations in oceanographic conditions (Figure 3a). BP was significantly correlated with Chl-a at the ASP sites ( $r^2 = 0.47$ , n = 15, p < 0.01) and WG sites ( $r^2 = 0.27$ , n = 15 p < 0.05). However, there was no significant correlation between BP and Chl-a in the EG (p > 0.05). In addition, BP at the EG site accounted for 48.9 ± 8.94% on average (range, 39.5%–49.8%) of primary production (PP) in MLD (Lim et al., 2019), which was higher than that estimated for the WG (average, 10.2 ± 4.98%; range, 7.11%–13.9%) and ASP (average, 9.15 ± 8.59%; range, 4.09%–19.1%) (Table 1).

Considering the median bacterial growth efficiency (BGE) of 11% (2%-28%) in the ASP (Williams et al., 2016), the unusually high BP:PP ratio (0.49 on average) in the EG implies that bacterial carbon demand (BCD = BP/BGE; 105  $\pm$  32.9 mmol C m<sup>-2</sup> d<sup>-1</sup>; Table S2 in Supporting Information S1) to support metabolic activities (production and respiration) is greater than the organic carbon produced through PP ( $22.5 \pm 11.78$  mmol C  $m^{-2} d^{-1}$  (Lim et al., 2019). The positive relationship between BP and the FWF in the EG ( $r^2 = 0.35$ , n = 16, p < 0.01; Figure 3b), along with the absence of significant relationship between BP and Chl-a, strongly suggests that glacial meltwater rather than phytoplankton provides an extra source of DOC to support the high BP in the EG (Figure 4a). Indeed, the concentration of DOC was significantly greater in the EG, where warmer glacier meltwater upwells to the surface mixed layer, compared with the WG and ASP (Figure S2b in Supporting Information S1). Wadham et al. (2019) reported that the potential DOC flux for total organic carbon emissions from Antarctic subglacial meltwater is estimated to be approximately 0.17 Tg C year<sup>-1</sup>, suggesting that significant amounts of fresh and labile DOC may be supplied to the surface mixed layer. Previous studies of the Greenland Ice Sheet have shown that DOC supplied through the export of meltwater is highly bioavailable and has a significant impact on microbial productivity and respiration (Bhatia et al., 2013; Lawson et al., 2014). Given that the Antarctic glacial ice shelves are considered a massive reservoir of organic carbon (ca. 5.5 Pg C) (Wadham et al., 2019), our results strongly imply that increases in glacial meltwater induced by global warming stimulate heterotrophic bacterial respiration that metabolizes DOM to CO<sub>2</sub>, reducing the efficiency of carbon sequestration in Antarctic coastal waters.



Figure 3. Relationships between bacterial production (BP) and chlorophyll-a (Chl-a) (a), and between BP and the freshwater fraction (b).





**Figure 4.** Schematic diagrams of bacterial responses to the dissolved organic carbon (DOC) derived from glacier meltwater in the eastern Getz (EG) Ice Shelf (panel a) and to the DOC derived from phytoplankton in the sea-ice melting system of the western Getz (WG) Ice Shelf (panel b) in the Amundsen Sea. Note that the EG region (glacier melting system) is characterized by an average deep-water depth of 817 m, which results in the direct contact of basal glacier with warmer modified circumpolar deep water (CDW), whereas the WG (sea-ice meltwater dominating system) features a relatively shallow water depth (566 m on average), which prevents warm CDW from melting the basal glacier. Consequently, the glacier-melting-dominated system (EG) features greatly enhanced bacterial production (BP) that is uncoupled with phytoplankton. In contrast, the sea-ice melting system (WG) a features high phytoplankton biomass resulting from shallow mixed-layer depth, and moderately enhanced BP coupled to phytoplankton.

#### 3.4. Bacterial Production Coupled With Phytoplankton in the Sea-Ice Melting System

There was a positive relationship between BP and phytoplankton in the WG (Figure 3a), but no significant correlation with the FWF (Figure 3b). The tight coupling between BP and Chl-a in the WG indicates that iron-containing meltwater from sea ice stimulates phytoplankton growth in a relatively stratified surface water column (Grotti et al., 2005; Lannuzel et al., 2008, 2010; Wang et al., 2014), which provides DOC to support BP (Figure 4b). Similarly, the tight coupling between BP and Chl-a in the ASP is consistent with findings reported in previous studies showing that phytoplankton are a major source of DOC supporting BP (Figure 3a) (Hyun et al., 2016; Williams et al., 2016). The positive relationship between BP and FWF in the ASP indicates that BP is supported in part by DOC released from the surrounding sea-ice meltwater (Dinasquet et al., 2017; Sipler & Connelly, 2015) (Figure 3b). When sea-ice meltwater was experimentally added to sub-AASWs, the BP increased significantly and 63% of the DOC was consumed by bacteria (Kähler et al., 1997). These results suggest that DOC released from the surrounding sea-ice meltwater, together with phytoplankton, supports the metabolic activity of heterotrophic bacteria in the sea-ice melting system of the ASP (Figures 3a and 3b).

19448007, 2022,

19, Downloaded

from

doi/10.1029/2021GL097627 by Hanyang University Library.

, Wiley

Jorary

on [24/07/2023]

### 4. Summary

Given that the Antarctic glacial ice shelves retain a large amount of organic carbon (Wadham et al., 2019), the enhanced BP and high BP:PP ratio on the eastern GtzIS, where glacier meltwater upwells into the surface layer, have critical implications for the impact of warming-induced glacier meltwater on carbon cycles in Antarctic waters. An increase in glacial meltwater could stimulate heterotrophic bacterial respiration that metabolizes DOM to  $CO_2$ , ultimately weakening carbon sequestration in the Antarctic coastal waters. Our results raise additional questions about the glacial melting-induced supply of DOC and its effects on the net metabolic balance between heterotrophic and autotrophic processes in Antarctic coastal waters.

# **Data Availability Statement**

The primary production data in Table 1 are available from Lim et al. (2019). The physico-chemical and bacterial parameters used in this study were uploaded to the Korea Polar Data Center (https://dx.doi.org/doi:10.22663/KOPRI-KPDC-00001893.2). The data of the current manuscript is provided by Korea Polar Data Center.

#### References

- Alderkamp, A. C., Mills, M. M., van Dijken, G. L., Laan, P., Thuróczy, C. E., Gerringa, L. J. A., et al. (2012). Iron from melting glaciers fuels phytoplankton blooms in the Amundsen Sea (Southern Ocean): Phytoplankton characteristics and productivity. *Deep-Sea Research Part II Topical Studies in Oceanography*, 71–76, 32–48. https://doi.org/10.1016/j.dsr2.2012.03.005
- Arrigo, K. R., van Dijken, G., & Long, M. (2008). Coastal Southern Ocean: A strong anthropogenic CO<sub>2</sub> sink. *Geophysical Research Letters*, 35(21), L21602. https://doi.org/10.1029/2008GL035624
- Arrigo, K. R., van Dijken, G. L., Castelao, R. M., Luo, H., Rennermalm, Å. K., Tedesco, M., et al. (2017). Melting glaciers stimulate large summer phytoplankton blooms in southwest Greenland waters. Geophysical Research Letters, 44(12), 6278–6285. https://doi. org/10.1002/2017GL073583
- Azam, F., Fenchel, T., Field, J., Gray, J., Meyer-Reil, L., & Thingstad, F. (1983). The ecological role of water-column microbes in the sea. *Marine Ecology Progress Series*, 10(13), 257–263. https://doi.org/10.3354/meps010257
- Azam, F., & Malfatti, F. (2007). Microbial structuring of marine ecosystems. Nature Reviews Microbiology, 5(10), 782–791. https://doi. org/10.1038/nrmicro1747
- Bhatia, M. P., Das, S. B., Xu, L., Charette, M. A., Wadham, J. L., & Kujawinski, E. B. (2013). Organic carbon export from the Greenland ice sheet. Geochimica et Cosmochimica Acta, 109, 329–344. https://doi.org/10.1016/j.gca.2013.02.006
- Brown, M. S., Munro, D. R., Feehan, C. J., Sweeney, C., Ducklow, H. W., & Schofield, O. M. (2019). Enhanced oceanic CO<sub>2</sub> uptake along the rapidly changing West Antarctic Peninsula. *Nature Climate Change*, 9(9), 678–683. https://doi.org/10.1038/s41558-019-0552-3
- Carlson, C. A., Hansell, D. A., Peltzer, E. T., & Smith, J. (2000). Stocks and dynamics of dissolved and particulate organic matter in the Southern Ross Sea, Antarctica. *Deep-Sea Research Part II: Topical Studies in Oceanography*, 47(15–16), 3201–3225. https://doi.org/10.1016/S0967-0645(00)00065-5
- Carlson, C. A., & Hansell, D. L. (2015). DOM source, sinks, reactivity, and budgets. In D. L. Hansell & C. A. (2nd Carlson) (Eds.), Biogeochemistry of marine dissolved organic matter (pp. 65–126). Academic Press.
- Chisholm, S. W., Falkowski, P. G., & Cullen, J. J. (2001). Dis-crediting ocean fertilization. Science, 294(5541), 309–310. https://doi.org/10.1126/ science.1065349
- del Giorgio, P. A., Cole, J. J., & Cimbleris, A. (1997). Respiration rates in bacteria exceed phytoplankton production in unproductive aquatic systems. Nature, 385(6612), 148–151. https://doi.org/10.1038/385148a0
- Delmont, T. O., Hammar, K. M., Ducklow, H. W., Yager, P. L., & Post, A. F. (2014). Phaeocystis Antarctica blooms strongly influence bacterial community structures in the Amundsen Sea polynya. *Frontiers in Microbiology*, 5(DEC), 1–13. https://doi.org/10.3389/fmicb.2014.00646
- Dinasquet, J., Richert, I., Logares, R., Yager, P., Bertilsson, S., & Riemann, L. (2017). Mixing of water masses caused by a drifting iceberg affects bacterial activity, community composition and substrate utilization capability in the Southern Ocean. *Environmental Microbiology*, 19(6), 2453–2467. https://doi.org/10.1111/1462-2920.13769
- Duarte, C. M., & Prairie, Y. T. (2005). Prevalence of heterotrophy and atmospheric CO<sub>2</sub> emissions from aquatic ecosystems. *Ecosystems*, 8(7), 862–870. https://doi.org/10.1007/s10021-005-0177-4
- Ducklow, H. W., Baker, K., Martinson, D. G., Quetin, L. B., Ross, R. M., Smith, R. C., et al. (2007). Marine pelagic ecosystems: The West Antarctic Peninsula. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 362(1477), 67–94. https://doi.org/10.1098/ rstb.2006.1955
- Ducklow, H. W., Purdie, D. A., Williams, P. J. L. B., & Davies, J. M. (1986). Bacterioplankton: A sink for carbon in a coastal marine plankton community. *Science*, 232(4752), 871–873. https://doi.org/10.1126/science.232.4752.865
- Ducklow, H. W., Schofield, O., Vernet, M., Stammerjohn, S., & Erickson, M. (2012). Multiscale control of bacterial production by phytoplankton dynamics and sea ice along the Western Antarctic Peninsula: A regional and decadal investigation. *Journal of Marine Systems*, 98(99), 26–39. https://doi.org/10.1016/j.jmarsys.2012.03.003
- Ducklow, H. W., Wilson, S. E., Post, A. F., Stammerjohn, S. E., Erickson, M., Lee, S., et al. (2015). Particle flux on the continental shelf in the Amundsen Sea Polynya and Western Antarctic Peninsula. *Elementa: Science of the Anthropocene*, 3. https://doi.org/10.12952/journal. elementa.000046
- Fang, L., Lee, S., Lee, S.-A., Hahm, D., Kim, G., Druffel, E. R. M., & Hwang, J. (2020). Removal of refractory dissolved organic carbon in the Amundsen Sea, Antarctica. Scientific Reports, 10(1), 1213. https://doi.org/10.1038/s41598-020-57870-6
- Grotti, M., Soggia, F., Ianni, C., & Frache, R. (2005). Trace metals distributions in coastal sea ice of Terra Nova Bay, Ross Sea, Antarctica. Antarctic Science, 17(2), 289–300. https://doi.org/10.1017/S0954102005002695
- Hansell, D. L., & Carlson, C. A. (2015). Biogeochemistry of marine dissolved organic matter (2nd Eds.), Academic Press.

#### Acknowledgments

This study was supported by the "Microbially mediated ecological and biogeochemical process study in the Arctic Ocean" funded by the Korean Ministry of Oceans and Fisheries through the Open Innovation Program (20220554), Korea Polar Research Institute (KOPRI, PE21110), and a grant from the National Research Foundation of Korea (NRF-2018R1A2B2006340).

- Hood, E., Battin, T. J., Fellman, J., O'Neel, S., & Spencer, R. G. M. (2015). Storage and release of organic carbon from glaciers and ice sheets. *Nature Geoscience*, 8(2), 91–96. https://doi.org/10.1038/ngeo2331
- Hopwood, M. J., Carroll, D., Browning, T. J., Meire, L., Mortensen, J., Krisch, S., & Achterberg, E. P. (2018). Non-linear response of summertime marine productivity to increased meltwater discharge around Greenland. *Nature Communications*, 9(1), 3256. https://doi.org/10.1038/ s41467-018-05488-8
- Hopwood, M. J., Carroll, D., Dunse, T., Hodson, A., Holding, J. M., Iriarte, J. L., et al. (2020). Review article: How does glacier discharge affect marine biogeochemistry and primary production in the Arctic? *The Cryosphere*, 14(4), 1347–1383. https://doi.org/10.5194/tc-14-1347-2020
- Hyun, J.-H., Kim, S.-H., Yang, E. J., Choi, A., & Lee, S. H. (2016). Biomass, production, and control of heterotrophic bacterioplankton during a late phytoplankton bloom in the Amundsen Sea Polynya, Antarctica. *Deep Sea Research Part II: Topical Studies in Oceanography*, 123, 102–112. https://doi.org/10.1016/j.dsr2.2015.10.001
- Jacobs, S., Jenkins, A., Hellmer, H., Giulivi, C., Nitsche, F., Huber, B., & Guerrero, R. (2012). The Amundsen Sea and the Antarctic ice sheet. Oceanography, 25(3), 154–163. https://doi.org/10.5670/oceanog.2012.90
- Jacobs, S. S., Jenkins, A., Giulivi, C. F., & Dutrieux, P. (2011). Stronger ocean circulation and increased melting under Pine Island Glacier ice shelf. *Nature Geoscience*, 4(8), 519–523. https://doi.org/10.1038/ngeo1188
- Jenkins, A., Dutrieux, P., Jacobs, S. S., McPhail, S. D., Perrett, J. R., Webb, A. T., & White, D. (2010). Observations beneath Pine Island Glacier in West-Antarctica and implications for its retreat. *Nature Geoscience*, 3(7), 468–472. https://doi.org/10.1038/ngeo890
- Jourdain, N. C., Mathiot, P., Merino, N., Durand, G., Le Sommer, J., Spence, P., et al. (2017). Ocean circulation and sea-ice thinning induced by melting ice shelves in the Amundsen Sea. Journal of Geophysical Research: Oceans, 122(3), 2550–2573. https://doi.org/10.1002/2016JC012509
- Kähler, P., Bjornsen, P. K., Lochte, K., & Antia, A. (1997). Dissolved organic matter and its utilization by bacteria during spring in the Southern Ocean. Deep Sea Research Part II: Topical Studies in Oceanography, 44(1–2), 341–353. https://doi.org/10.1016/S0967-0645(96)00071-9
  Karl, D. M. (2007). Microbial oceanography: Paradigms, processes and promise. Nature Reviews Microbiology, 5(10), 759–769. https://doi.
- org/10.1038/nrmicro1749 Kim, B., Kim, S.-H., Min, J.-O., Lee, Y., Jung, J., Kim, T.-W., et al. (2022). Bacterial metabolic response to change in phytoplankton communi-
- Kini, S., Kini, S.-H., Min, J.-O., Lee, T., Jung, J., Kini, T.-W., et al. (2022). Bacterial inetabolic response to change in phytoplankton communities and resultant effects on carbon cycles in the Amundsen Sea Polynya, Antarctica. *Frontiers in Marine Science*, 9(June), 1–15. https://doi. org/10.3389/fmars.2022.872052
- Kim, B. K., Lee, S. H., Ha, S. Y., Jung, J., Kim, T. W., Yang, E. J., et al. (2018). Vertical distributions of macromolecular composition of particulate organic matter in the water column of the Amundsen Sea Polynya during the summer in 2014. *Journal of Geophysical Research: Oceans*, 123(2), 1393–1405. https://doi.org/10.1002/2017JC013457
- Kim, H., & Ducklow, H. W. (2016). A decadal (2002–2014) analysis for dynamics of heterotrophic bacteria in an Antarctic coastal ecosystem: Variability and physical and biogeochemical forcings. *Frontiers in Marine Science*, 3(NOV), 1–18. https://doi.org/10.3389/fmars.2016.00214
- Kim, I., Hahm, D., Rhee, T. S., Kim, T. W., Kim, C., & Lee, S. (2016). The distribution of glacial meltwater in the Amundsen Sea, Antarctica, revealed by dissolved helium and neon. Journal of Geophysical Research: Oceans, 121(3), 1654–1666. https://doi.org/10.1002/2015JC011211
- Kim, J.-G., Park, S.-J., Quan, Z.-X., Jung, M.-Y., Cha, I.-T., Kim, S.-J., et al. (2014). Unveiling abundance and distribution of planktonic bacteria and Archaea in a polynya in Amundsen Sea, Antarctica. Environmental Microbiology, 16(6), 1566–1578. https://doi.org/10.1111/1462-2920.12287
- Kim, S.-J., Kim, J.-G., Lee, S.-H., Park, S.-J., Gwak, J.-H., Jung, M.-Y., et al. (2019). Correction to: Genomic and metatranscriptomic analyses of carbon remineralization in an Antarctic polynya. *Microbiome*, 7(1), 38. https://doi.org/10.1186/s40168-019-0655-0
- Kirchman, D. L. (2008). Introduction and overview. In D. L. Kirchman (Ed.), Microbial ecology of the oceans (pp. 1–26). Wiley-Blackwell.
- Lannuzel, D., Schoemann, V., de Jong, J., Chou, L., Delille, B., Becquevort, S., & Tison, J. L. (2008). Iron study during a time series in the Western Weddell pack ice. *Marine Chemistry*, 108(1–2), 85–95. https://doi.org/10.1016/j.marchem.2007.10.006
- Lannuzel, D., Schoemann, V., De Jong, J., Pasquer, B., Van Der Merwe, P., Masson, F., et al. (2010). Distribution of dissolved iron in Antarctic sea ice: Spatial. seasonal. and inter-annual variability. *Journal of Geophysical Research*, 115(3), 1–13. https://doi.org/10.1029/2009JG001031
- Lawson, E. C., Wadham, J. L., Tranter, M., Stibal, M., Lis, G. P., Butler, C. E. H., et al. (2014). Greenland ice sheet exports labile organic carbon to the arctic oceans. *Biogeosciences*, 11(14), 4015–4028. https://doi.org/10.5194/bg-11-4015-2014
- Lee, S., Hwang, J., Ducklow, H. W., Hahm, D., Lee, S. H., Kim, D., et al. (2017). Evidence of minimal carbon sequestration in the productive Amundsen Sea Polynya. *Geophysical Research Letters*, 44(15), 7892–7899. https://doi.org/10.1002/2017GL074646
- Lee, Y., Yang, E. J., Park, J., Jung, J., Kim, T. W., & Lee, S. H. (2016). Physical-biological coupling in the Amundsen Sea, Antarctica: Influence of physical factors on phytoplankton community structure and biomass. *Deep-Sea Research Part I Oceanographic Research Papers*, 117(October), 51–60. https://doi.org/10.1016/j.dsr.2016.10.001
- Legendre, L., & Le Fèvre, J. (1995). Microbial food webs and the export of biogenic carbon in oceans. Aquatic Microbial Ecology, 9(1), 69–77. https://doi.org/10.3354/ame009069
- Lim, Y. J., Kim, T. W., Lee, S. H., Lee, D., Park, J., Kim, B. K., et al. (2019). Seasonal variations in the Small phytoplankton contribution to the total primary production in the Amundsen Sea, Antarctica. *Journal of Geophysical Research: Oceans*, 124(11), 8324–8341. https://doi. org/10.1029/2019JC015305
- Lønborg, C., Álvarez–Salgado, X. A., Letscher, R. T., & Hansell, D. A. (2018). Large stimulation of recalcitrant dissolved organic carbon degradation by increasing ocean temperatures. Frontiers in Marine Science, 4(JAN), 1–11. https://doi.org/10.3389/fmars.2017.00436
- Lønborg, C., Baltar, F., Carreira, C., & Morán, X. A. G. (2019). Dissolved organic carbon source influences Tropical coastal heterotrophic bacterioplankton response to experimental warming. *Frontiers in Microbiology*, 10(December). https://doi.org/10.3389/fmicb.2019.02807
- Mankoff, K. D., Jacobs, S. S., Tulaczyk, S. M., & Stammerjohn, S. E. (2012). The role of pine island glacier ice shelf basal channels in deep-water upwelling, polynyas and ocean circulation in pine island bay, Antarctica. Annals of Glaciology, 53(60), 123–128. https://doi. org/10.3189/2012AoG60A062
- Montes-Hugo, M., Doney, S. C., Ducklow, H. W., Fraser, W., Martinson, D., Stammerjohn, S. E., & Schofield, O. (2009). Recent changes in phytoplankton communities associated with rapid regional climate change along the Western Antarctic Peninsula. *Science*, 323(5920), 1470–1473. https://doi.org/10.1126/science.1164533
- Morán, X. A. G., Gasol, J. M., Pedrós-Alió, C., & Estrada, M. (2001). Dissolved and particulate primary production and bacterial production in offshore Antarctic waters during austral summer: Coupled or uncoupled? *Marine Ecology Progress Series*, 222, 25–39. https://doi. org/10.3354/meps222025
- Morlighem, M., Rignot, E., Binder, T., Blankenship, D., Drews, R., Eagles, G., et al. (2020). Deep glacial troughs and stabilizing ridges unveiled beneath the margins of the Antarctic ice sheet. *Nature Geoscience*, 13(2), 132–137. https://doi.org/10.1038/s41561-019-0510-8
- Musilova, M., Tranter, M., Wadham, J., Telling, J., Tedstone, A., & Anesio, A. M. (2017). Microbially driven export of labile organic carbon from the Greenland ice sheet. *Nature Geoscience*, 10(5), 360–365. https://doi.org/10.1038/ngeo2920
- Nagata, T. (2008). Organic matter-bacteria interactions in seawater. In D. L. Kirchman (Ed.), *Microbial ecology of the oceans* (pp. 207–241). Wiley-Blackwell.

- Nakayama, Y., Timmermann, R., Rodehacke, C. B., Schröder, M., & Hellmer, H. H. (2014). Modeling the spreading of glacial meltwater from the Amundsen and Bellingshausen seas. *Geophysical Research Letters*, *41*(22), 7942–7949. https://doi.org/10.1002/2014GL061600
- Nicholes, M. J., Williamson, C. J., Tranter, M., Holland, A., Poniecka, E., Yallop, M. L., & Anesio, A. (2019). Bacterial dynamics in supraglacial habitats of the Greenland ice sheet. *Frontiers in Microbiology*, 10(JUL). https://doi.org/10.3389/fmicb.2019.01366
- Ortega-Retuerta, E., Reche, I., Pulido-Villena, E., Agustí, S., & Duarte, C. M. (2008). Exploring the relationship between active bacterioplankton and phytoplankton in the Southern Ocean. Aquatic Microbial Ecology, 52(1), 99–106. https://doi.org/10.3354/ame01216
- Paulsen, M. L., Nielsen, S. E. B., Müller, O., Møller, E. F., Stedmon, C. A., Juul-Pedersen, T., et al. (2017). Carbon Bioavailability in a high Arctic Fjord influenced by glacial meltwater, NE Greenland. Frontiers in Marine Science, 4(JUN), 1–19. https://doi.org/10.3389/fmars.2017.00176
- Raiswell, R., Tranter, M., Benning, L. G., Siegert, M., De'ath, R., Huybrechts, P., & Payne, T. (2006). Contributions from glacially derived sediment to the global iron (oxyhydr)oxide cycle: Implications for iron delivery to the oceans. *Geochimica et Cosmochimica Acta*, 70(11), 2765–2780. https://doi.org/10.1016/j.gca.2005.12.027
- Randall-Goodwin, E., Meredith, M. P., Jenkins, A., Yager, P. L., Sherrell, R. M., Abrahamsen, E. P., et al. (2015). Freshwater distributions and water mass structure in the Amundsen Sea Polynya region, Antarctica. *Elementa*, *3*, 1–22. https://doi.org/10.12952/journal.elementa.000065 Rignot, E., Jacobs, S., Mouginot, J., & Scheuchl, B. (2013). Ice-shelf melting around Antarctica. *Science*, *341*(6143), 266–270. https://doi.
  - org/10.1126/science.1235798
- Rignot, E., Mouginot, J., Scheuchl, B., Van Den Broeke, M., Van Wessem, M. J., & Morlighem, M. (2019). Four decades of Antarctic ice sheet mass balance from 1979–2017. Proceedings of the National Academy of Sciences of the United States of America, 116(4), 1095–1103. https:// doi.org/10.1073/pnas.1812883116
- Silvano, A., Rintoul, S. R., Peña-Molino, B., Hobbs, W. R., Van Wijk, E., Aoki, S., et al. (2018). Freshening by glacial meltwater enhances
- melting of ice shelves and reduces formation of Antarctic Bottom Water. *Science Advances*, 4(4), 1–12. https://doi.org/10.1126/sciadv.aap9467 Sipler, R. E., & Connelly, T. L. (2015). Bioavailability of surface dissolved organic matter to aphotic bacterial communities in the Amundsen Sea

Polynya, Antarctica. *Elementa*, *3*, 1–13. https://doi.org/10.12952/journal.elementa.000060

- Smith, H. J., Foster, R. A., McKnight, D. M., Lisle, J. T., Littmann, S., Kuypers, M. M. M., & Foreman, C. M. (2017). Microbial formation of labile organic carbon in Antarctic glacial environments. *Nature Geoscience*, 10(5), 356–359. https://doi.org/10.1038/ngeo2925
- Takahashi, T., Sutherland, S. C., Wanninkhof, R., Sweeney, C., Feely, R. A., Chipman, D. W., et al. (2009). Climatological mean and decadal change in surface ocean pCO<sub>2</sub>, and net sea–air CO<sub>2</sub> flux over the global oceans. *Deep Sea Research Part II: Topical Studies in Oceanography*, 56(8–10), 554–577. https://doi.org/10.1016/j.dsr2.2008.12.009
- Turner, J., Barrand, N. E., Bracegirdle, T. J., Convey, P., Hodgson, D. A., Jarvis, M., et al. (2014). Antarctic climate change and the environment: An update. *Polar Record*, 50(3), 237–259. https://doi.org/10.1017/S0032247413000296
- Wadham, J. L., Hawkings, J. R., Tarasov, L., Gregoire, L. J., Spencer, R. G. M., Gutjahr, M., et al. (2019). Ice sheets matter for the global carbon cycle. Nature Communications, 10(1), 3567. https://doi.org/10.1038/s41467-019-11394-4
- Wang, S., Bailey, D., Lindsay, K., Moore, J. K., & Holland, M. (2014). Impact of sea ice on the marine iron cycle and phytoplankton productivity. *Biogeosciences*, 11(17), 4713–4731. https://doi.org/10.5194/bg-11-4713-2014
- Wei, W., Blankenship, D. D., Greenbaum, J. S., Gourmelen, N., Dow, C. F., Richter, T. G., et al. (2020). Getz ice shelf melt enhanced by freshwater discharge from beneath the west Antarctic ice sheet. *The Cryosphere*, 14(4), 1399–1408. https://doi.org/10.5194/tc-14-1399-2020
- Williams, C. M., Dupont, A. M., Loevenich, J., Post, A. F., Dinasquet, J., & Yager, P. L. (2016). Pelagic microbial heterotrophy in response to a highly productive bloom of Phaeocystis Antarctica in the Amundsen Sea Polynya, Antarctica. *Elementa*, 4, 1–18. https://doi.org/10.12952/ journal.elementa.000102
- Yager, P. L., Sherrell, R. M., Stammerjohn, S. E., Alderkamp, A. C., Schofield, O., Abrahamsen, E. P., et al. (2012). ASPIRE: The Amundsen Sea Polynya international research expedition. *Oceanography*, 25(3), 40–53. https://doi.org/10.5670/oceanog.2012.73

## **References From the Supporting Information**

- Brainerd, K. E., & Gregg, M. C. (1995). Surface mixed and mixing layer depths. Deep-Sea Research Part I: Oceanographic Research Papers, 42(9), 1521–1543. https://doi.org/10.1016/0967-0637(95)00068-H
- Catalano, G., Povero, P., Fabiano, M., Benedetti, F., & Goffart, A. (1997). Nutrient utilisation and particulate organic matter changes during summer in the upper mixed layer (Ross Sea, Antarctica). Deep Sea Research Part I: Oceanographic Research Papers, 44(1), 97–112. https:// doi.org/10.1016/S0967-0637(97)85248-X
- Fukuda, R., Ogawa, H., Nagata, T., & Koike, I. (1998). Direct determination of carbon and nitrogen contents of natural bacterial assemblages in marine environments. *Applied and Environmental Microbiology*, 64(9), 3352–3358. https://doi.org/10.1128/AEM.64.9.3352-3358.1998
- Kirchman, D., K'nees, E., & Hodson, R. (1985). Leucine incorporation and its potential as a measure of protein synthesis by bacteria in natural aquatic systems. Applied and Environmental Microbiology, 49(3), 599–607. https://doi.org/10.1128/aem.49.3.599-607.1985
- Parsons, T. R., Maita, Y., & Lalli, C. M. (1984). A manual of chemical and biological methods for seawater analysis (p. 173). Pergamon Press. Porter, K. G., & Feig, Y. S. (1980). The use of DAPI for identifying and counting aquatic microflora1. *Limnology & Oceanography*, 25(5), 943–948. https://doi.org/10.4319/10.1980.25.5.0943
- Rivaro, P., Luisa Abelmoschi, M., Grotti, M., Ianni, C., Magi, E., Margiotta, F., et al. (2012). Combined effects of hydrographic structure and iron and copper availability on the phytoplankton growth in Terra Nova Bay Polynya (Ross Sea, Antarctica). Deep-Sea Research Part I Oceanographic Research Papers, 62, 97–110. https://doi.org/10.1016/j.dsr.2011.12.008
- Simon, M., & Azam, F. (1989). Protein content and protein synthesis rates of planktonic marine bacteria. Marine Ecology Progress Series, 51, 201–213. https://doi.org/10.3354/meps051201